Author’s response to reviews

Title: Patients with Glycogen Storage Diseases Undergoing Anesthesia: A Case Series

Authors:
Carmelina Gurrieri (Gurrieri.Carmelina@mayo.edu)
Juraj Sprung (sprung.juraj@mayo.edu)
Toby Weingarten (weingarten.toby@mayo.edu)
Mary Warner (warner.mary@mayo.edu)

Version: 2 Date: 31 Aug 2017

Author’s response to reviews:

BANE-D-17-00182R1

Patients with Glycogen Storage Diseases Undergoing Anesthesia: A Case Series
Carmelina Gurrieri, MD; Juraj Sprung, MD, PhD; Toby Weingarten; Mary E Warner, MD

BMC Anesthesiology

Dear Dr. Riha,

Thank you for the provisional acceptance of our manuscript "Patients with Glycogen Storage Diseases Undergoing Anesthesia: A Case Series" (BANE-D-17-00182R1). The reviewer asked that we provide a more clear definition of metabolic acidosis in the Methods section. We have provided this. Please let us know if you or the reviewers have any further questions.

Sincerely yours,

Toby Weingarten
Minor comment

I would suggest to re-phrase the definition of metabolic acidosis in the Methods section (... primarily secondary ...?) to make it more clear for international readers of the journal.

RESPONSE:

We have revised this to be clearer, as follows: “metabolic acidosis was defined as pH imbalance due to a reduction in bicarbonate (HCO3−) of 22 mmol/L or lower, which was not a result from secondary causes, e.g., chronic respiratory alkalosis”